Author's response to reviews

Title:Differences in muscle activity during hand dexterity tasks between women with arthritis and a healthy reference group

Authors:

Sofia Brorsson (sbo@du.se)
Anna Nilsdotter (Anna.Nilsdotter@regionhalland.se)
Carina Thorstensson (carina.thorstensson@registercentrum.se)
Ann Bremander (ann.bremander@spenshult.se)

Version:5Date:10 April 2014

Author's response to reviews:

Dear editor!

I have now made the changes concerning:

- 1) By way of a section ?Acknowledgements? We have added thanks and sources
- 2. Please reformat the title page at the front of your manuscript file. DONE
- 3. page 11/12: the sentence "Furthermore, the force used was at a considerably lower percentage relative to their maximum force than when the tests were performed by patients with arthritis." is repeated twice. DONE

All the best Sofia Brorsson

3)